Author's response to reviews

Title: Myocardial Bridging as a Cause of Acute Myocardial Infarction: A Case Report

Authors:

Ramazan Akdemir (rakdemir@yahoo.com)
Huseyin Gunduz (huseyingunduz@hotmail.com)
Yunus Emiroglu (yemiroglu@yahoo.com)
Cihangir Uyan (cihangirayten@superonline.com)

Version: 4 Date: 9 Aug 2002

1-The quality of written english: I did my best.
2-The ECG accompanied the report.
3-We decided primary percutaneous coronary intervention, but the lesion was not suitable for PTCA/stent replacement.
4-we explained that profound anemia precipitated acute st-elevation MI in the presence of myocardial bridging by smoking induced endothelial dysfunction and coronary thrombosis.
5-Discussion section of the manuscript was extended.